Author's response to reviews

Title: Obatoclax is a Direct and Potent Antagonist of Membrane-restricted Mcl-1 and is Synthetic Lethal with Treatment that Induces Bim

Authors:

Mai Nguyen (mai.nguyen@mcgill.ca)
Regina Cencic (regina.cencic@mcgill.ca)
Franziska Ertel (franzi.ertel@gmx.de)
Cynthia Bernier (cynthia.bernier@mcgill.ca)
Jerry Pelletier (jerry.pelletier@mcgill.ca)
Anne Roulston (anne.roulston@mcgill.ca)
John R Silvius (john.silvius@mcgill.ca)
Gordon C Shore (gordon.shore@mcgill.ca)

Version: 2
Date: 16 March 2015

Author's response to reviews:

Please find the revised manuscript MS: 2145487686163838 to include:

1) Line and page numbers.

2) e-mail addresses of each co-author on the title page.

3) A conclusion section.

Thank you

Gordon Shore